Author's response to reviews

Title: Glutathione S-Transferase L1 Multiplex Serology as a Measure of Cumulative Infection with Human Papillomavirus

Authors:

- Hilary A Robbins (hilary.robbins@nih.gov)
- Yan Li (yli@survey.umd.edu)
- Carolina Porras (cporras@proyectoguanacaste.org)
- Michael Pawlita (m.pawlita@dkfz-heidelberg.de)
- Arpita Ghosh (arpita.ghosh@nih.gov)
- Ana Cecilia Rodriguez (acrodriguez@racsa.co.cr)
- Mark Schiffman (schiffmm@mail.nih.gov)
- Sholom Wacholder (wacholds@mail.nih.gov)
- Troy J Kemp (kemptj@mail.nih.gov)
- Paula Gonzalez (GonzalezP@fellow.iarc.fr)
- John Schiller (schillej@mail.nih.gov)
- Douglas Lowy (lowyd@mail.nih.gov)
- Mark Esser (EsserM@MedImmune.com)
- Katie Matys (Katie.Matys@ppdi.com)
- Wim Quint (w.g.v.quint@ddl.nl)
- Leen-Jan van Doorn (Leen-Jan.van.Doorn@ddl.nl)
- Rolando Herrero (HerreroR@iarc.fr)
- Ligia A Pinto (pintol@mail.nih.gov)
- Allan Hildesheime (hildesha@mail.nih.gov)
- Tim Waterboer (tiwabo@googlemail.com)
- Mahboobeh Safaeian (safaianm@mail.nih.gov)

Version: 2 Date: 22 November 2013

Author's response to reviews: see over
November 22, 2013

Philippa Harris
Executive Editor
BMC Infectious Diseases

Dear Dr. Harris,

We have revised our manuscript to include the changes requested by Sheryl Ramos in the BMC Infectious Diseases editorial office via e-mail on November 22, 2013.

First, we have included a list of e-mail addresses for all authors immediately after author affiliations.

To the methods we have added the following text: “The CVT protocol was approved by the institutional review boards of the U.S. National Cancer Institute and the Costa Rican INCIENSA, and all participants signed IRB-approved informed consent forms.”

There is now one section labeled “Acknowledgements” at the end of the manuscript.

Finally, to the “Funding” subheading under “Acknowledgements,” we have added the following text: “This study was designed, conducted, and written by NCI investigators with input from investigators at GlaxoSmithKline.”

Thank you for your consideration of our manuscript.

Sincerely,

Hilary A. Robbins, MSPH